Author's response to reviews

Title: The CHOICES/OPCIONES project: Pairing a patient decision aid and patient navigation to improve colorectal cancer screening in vulnerable populations

Authors:

Alison T Brenner (alison.brenner@unc.edu)
Christina M Getrich (chgetrich@salud.unm.edu)
Michael P Pignone (michael_pignone@med.unc.edu)
Robert Rhyne (rrhyne@salud.unm.edu)
Richard M Hoffman (rhoffman@unm.edu)
Andrew McWilliams (andrew.mcwilliams@carolinashealthcare.org)
Brisa Urquieta de Hernandez (brisa.hernandez@carolinashealthcare.edu)
Mark Weaver (mark_weaver@med.unc.edu)
Hazel Tapp (hazel.tapp@carolinashealthcare.org)
Khalil Harbi (harbi@email.unc.edu)
Daniel S Reuland (daniel_reuland@med.unc.edu)

Version: 5  Date: 1 May 2014

Author's response to reviews:

Dear Editors,

Thank you for the opportunity to re-submit our manuscript. I have addressed the formatting changes requested by the editorial board including:

1. Updating the title format to reflect the journal style for study protocol articles.
2. Including the date of trial registration in the abstract.
3. Including a statement in the methods section about obtaining informed consent from each participant.
4. The location of Trial Status.
5. The location of my funding statement.
6. The structure of my Author Contributions section.

We hope that you find this paper interesting and will consider its publication in Trials.

Sincerely,